

ANSWERS TO REVIEWERS



April 21st, 2014

Dear Editor,

Please find enclosed the edited manuscript in Word format (file name: 2429-review.doc).

Title: Utility of Arthroscopic Guided Synovial Biopsy in Understanding Synovial Tissue Pathology in Health and Disease States

Author: Mihir D Wechalekar, Malcolm D Smith

Name of Journal: *World Journal of Orthopaedics*

ESPS Manuscript NO: 9686

We thank the reviewers for their comments. The manuscript has been improved according to the suggestions of reviewers. Following the suggestion of the reviewer regarding additional detail on the technique of various biopsies, this has been reviewed by one of the authors of this paper elsewhere, and has now been appropriately mentioned and referenced.

The formatting has been updated and PMID numbers added to references and typesetting were corrected

Thank you again for publishing our manuscript in the *World Journal of Orthopaedics*.

Sincerely yours,

Dr Mihir D Wechalekar, MD, FRACP
Rheumatology Research Unit
Repatriation General Hospital
Daw Park SA 5041, Australia
Tel: +61 8 8275 1819, **Fax:** +61 8 8374 4276.
and Flinders University of South Australia, Australia

E-mail: mihir.wechalekar@health.sa.gov.au